Author's response to reviews

Title: The effect of coaching on the simulated malingering of memory impairment

Authors:

Jascha Rüsseler (jascha.ruesseler@nat.uni-magdeburg.de)
Alexandra Brett (alexandra.brett@ovgu.de)
Ulrike Klaue (ulrike.klaue@web.de)
Michael Sailer (michael.sailer@med.ovgu.de)
Thomas F. Münte (thomas.muente@med.ovgu.de)

Version: 6 Date: 29 August 2008

Author's response to reviews:

Action letter on the Ms:
MS: 1384429768189208
The effect of coaching on the simulated malingering of memory impairment
Jascha Rüsseler, Alexandra Brett, Ulrike Klaue, Michael Sailer and Thomas F. Münte

First, I want to thank the editors and reviewers for their work with the paper. We are happy that it is finally accepted for publication in BMC Neurology.

1. Please include the email addresses for all authors on the title page of your manuscript.
The email-addresses of all authors are now provided on the title page.

2. Can you please document the name of the ethical committee, which granted you permission to perform this investigation, in the Participants sub-section of your Methods section. Can you also document how participants' consent was collected and recorded (whether written or verbal).

We added the following statement in the participants section:
The study protocol was approved by the ethical committee of Magdeburg University. All participants gave written informed consent prior to psychometric testing.

We would also request that you go through the manuscript formatting checklist one more time and ensure that your revised manuscript conforms to all of the points.

We checked the manuscript once again and to the best of our knowledge it conforms to all points of the formatting checklist.